Author's response to reviews

Title: Standard Versus Prosocial Online Support Groups for Distressed Breast Cancer Survivors: A Randomized Controlled Trial

Authors:

Stephen J Lepore (slepore@temple.edu)
Joanne S Buzaglo (joanne@cancersupportcommunity.org)
Morton A Lieberman (Morton.Lieberman@ucsf.edu)
Mitch Golant (mitch@cancersupportcommunity.org)
Adam Davey (adavey@temple.edu)

Version: 2 Date: 19 August 2011

Author's response to reviews:

We addressed all revisions:
1. Fixed abstract structure
2. Fixed headings in paper. Note: There was no guidance on subheadings, so we used APA guidelines. Please modify if necessary.
3. Put abbreviations in paragraph format.
4. Improved Figure image
5. Removed additional files
6. Ran spell/grammar check and made minor editorial changes and reference updates

Thank you!